%------------------------------------------------------------------------------
% Here please write the date of submission of paper or its revisions:
%------------------------------------------------------------------------------
%
\documentclass[12pt, reqno]{amsart}
\usepackage{amsmath, amsthm, amscd, amsfonts, amssymb, graphicx, color}
\usepackage[bookmarksnumbered, colorlinks, plainpages]{hyperref}


\textheight 22.5truecm \textwidth 14.5truecm
\setlength{\oddsidemargin}{0.35in}\setlength{\evensidemargin}{0.35in}

\setlength{\topmargin}{-.5cm}

\newtheorem*{theoA}{Theorem A}
\newtheorem*{theoB}{Theorem B}
\newtheorem*{theoC}{Theorem C}
\newtheorem*{theoD}{Theorem D}
\newtheorem*{theoE}{Theorem E}
\newtheorem*{theoF}{Theorem F}
\newtheorem*{theoG}{Theorem G}
\newtheorem*{theoH}{Theorem H}
\newtheorem*{theoI}{Theorem I}
\newtheorem*{theoJ}{Theorem J}
\newtheorem*{theoK}{Theorem K}
\newtheorem*{theoL}{Theorem L}
\newtheorem{theo}{Theorem}[section]
\newtheorem{lem}{Lemma}[section]
\newtheorem{cor}{Corollary}[section]
\newtheorem{note}{Note}[section]
\newtheorem{exm}{Example}[section]
\newtheorem{defi}{Definition}[section]
\newtheorem{rem}{Remark}[section]
\newtheorem{obs}{Observation}[section]
\newtheorem{ques}{Question}[section]
\newcommand{\ol}{\overline}
\newcommand{\be}{\begin{equation}}
	\newcommand{\ee}{\end{equation}}
\newcommand{\beas}{\begin{eqnarray*}}
	\newcommand{\eeas}{\end{eqnarray*}}
\newcommand{\bea}{\begin{eqnarray}}
	\newcommand{\eea}{\end{eqnarray}}
\numberwithin{equation}{section}

\begin{document}
\begin{thebibliography}{99}
			
		\bibitem{Ban_Georgian_2008}A. Banerjee, {\it On the Uniqueness of Meromorphic functions that share two sets}, Georgian Math. J., 15(1)(2008), 21-38.
		\bibitem{Ban+Roy_Contemp_2023} A. Banerjee and A. Roy, {\it Uniqueness of a Meromorphic Function Partially and Normally Sharing Small Functions with Its Different Variants of Generalized Operator}, J. Contemp. Math. Anal., 58(1)(2023), 71-88.
		\bibitem{Chen+Xu_Houston_2017} S. Chen and W. C. Lin, {\it Periodicity and uniqueness of meromorphic functions with three shared values}, J. Math. Anal. Appl., 385(2012), 485-490.
		
		\bibitem{Chen+Huang_Contemp_2022} W. J. Chen and Z. G. Huang, {\it Uniqueness of Meromorphic Functions Concerning Their Derivatives and Shifts with Partially Shared Values,} J. Contemp. Math. Anal., 57(2022), 232–241.
	%	\bibitem{Ban+Roy_JIMS_2021} A. Banerjee, A. Roy, {\it Uniqueness of meromorphic functions ordinarily and partially sharing values with reduced linear $c$-shift operators}, J. Indian Math. Soc., 88(3-4)(2021), 201-216.
	%	\bibitem{Chen_CMFT_2018} S. Chen, {\it On uniqueness of meromorphic functions and their difference operators with partially shared valuers,} Comput. Methods Funct. Theory, 18(2018), 529-536.
	%	\bibitem{Chen+Yi_Resultsmath_2013}  Z. X. Chen, H. X. Yi, {\it On Sharing Values of Meromorphic Functions and Their Differences}, Results. Math., 63(2013), 557–565.
		\bibitem{Chiang+Feng_Ramanujan_2008} Y. M. Chiang and S. J. Feng, {\it On the Nevanlinna characteristic of $f(z+\eta)$ and difference equationsin the complex plane}, Ramanujan J., 16(2008), 105-129.
		\bibitem{Halburd+Korhonen_Tohge_TAMS_2014} R. G. Halburd, R. J. Korhonen and K. Tohge, {\it Holomorphic curves with shift invariant hyperplane preimage,} Trans. Am. Math. Soc., 366(2014), 4267-4298.
		\bibitem{Hayman_1964} W. K. Hayman, {\it Meromorphic Functions}, The Clarendon Press,  Oxford (1964).
		\bibitem{Heittokangas_Com. var_2011} J. Heittokangas, R. Korhonen, I. Laine and J.  Rieppo, {\it Uniqueness of meromorphic functions sharing values with their shifts}, Complex Var. Elliptic Equations, 56(1–4)(2010), 81–92.
		\bibitem{Heittokangas_JMAA_2009} J. Heittokangas, R. Korhonen, I. Laine, J.  Rieppo and J. Zhang, {\it Value sharing results for shifts of meromorphic function and sufficient conditions for periodicity}, J. Math. Anal. Appl., 355(2009), 352-363.
\bibitem{Lin+Ishizaki_JMAA_2018} W. C. Lin and K. Ishizaki, {\it A ``3IM+1CM" result for periodic meromorphic functions}, J. Math. Anal. Appl., 466(1)(2018), 726-732.
\bibitem{Li+Yi_Bull.Korean_2016} X. M. Li and H. X. Yi, {\it Meromorphic functions sharing four values with their difference operations or shifts,} Bull. Korean Math. Soc., 53(4)(2016), 1213-1235.
\bibitem{Pramanik+Sarkar_Math.Studii_2024} D. C. Pramanik and A. Sarkar, {\it Uniqueness of shift and derivatives of meromorphic functions}, Math. Studii., 61(2)(2024), 160-167.
	%	\bibitem{Gao_Archmath_2011} G. Li, {\it Entire functions sharing one or two finite values CM with their shifts or difference operators}, Arch. Math. (Basel), 97(2011), 475-483.
		%\bibitem{Li+Mei+Chen_Adv diff eq_2017} S. Li, D. Mei, B. Chen, {\it Uniqueness of meromorphic functions sharing two values with their difference operator}, Adv. Differ. Equ., 390(2017), 1-9.
		\bibitem{Qi_Ann polon_2011} X. Qi, {\it Value distribution and uniqueness of difference polynomials and entire solutions of difference equations}, Ann. Polon. Math., 102(2)(2011), 129-142.
		
	%	\bibitem{Qi+Yang_Anal math_2020} X. G. Qi, L. Z. Yang, {\it Meromorphic functions that share values with their shifts or their $n$-th order differences}, Anal.Math., 46(4)(2020), 843-865.
		\bibitem{Yang_Springer_1993} L. Yang, {\it Value distribution theory}, Springer, Berlin, 1993.
		\bibitem{Yi+Yang_2003_Sci.Press} H. X. Yi and C. C. Yang, {\it Uniqueness Theory of Meromorphic Functions}, Science Press, Beijing (2003).
		
		
		%\bibitem{Yi_bullaus_1995} H. X. Yi, {\it Unicity theorems for meromorphic or entire functions II}, Bull. Austral. Math. Soc., 52(2)(1995), 215-224.
		%\bibitem{Yi_1994}H. X. Yi, {\it Uniqueness of meromorphic functions and a question of Gross}, J. Sc. China (Ser. A), 37(7)(1994), 802-813.
		%\bibitem{Yi_1996} H. X .Yi, {\it Uniqueness theorems for meromorphic functions}, Chinese Ann. Math. (Ser. A), 17(6)(1996), 397-404.
		%\bibitem{Ban+Ahamed_Palermo_2018} A. Banerjee and M. B. Ahamed, {\it Polynomial of meromorphic function and its k-th derivative sharing a set}, Rend. Circ. Mat. Palermo. (Ser II), 67(3)(2018), 581-598.
		%\bibitem{Yang+hua_mat.fiz_1997} C. C. Yang and H. X. Hua, {\it Unique polynomials of entire and meromorphic functions}, Mat. Fiz. Anal. Geom., 4(3)(1997), 391-399. 
		%\bibitem{Li+Yang_1996} P. Li and C. C. Yang, {\it On the unique range sets for meromorphic functions}, Proc. Amer. Math. Soc., 124(1996), 177-185.
		%\bibitem{Li+Yang_kodai_1995} P. Li and C. C. Yang, {\it Some further results on the unique range sets of meromorphic functions}, Kodai Math. J., 18(1995), 437-450.
		%\bibitem{Gross_1982} F. Gross and C. C. Yang, {\it On pre-image and range sets of meromorphic functions}, Proc. Japan Acad., 58(1982),	17-20
		
		%\bibitem{Fujimoto_Amer.J.Math_2000} H. Fujimoto, {\it On uniqueness of meromorphic functions sharing finite sets}, Amer. J. Math., 122(6)(2000), 1175-1203.
		
		%\bibitem{Bartels_1998} S. Bartel, {\it Meromorphic functions sharing a set with 17 elements ignoring multiplicities,} Com. Var. Theo. Appl., 39(1)(1998), 85-92.
		%\bibitem{Ban_Nihonkai_2013} A. Banerjee, {\it Bi-Unique range sets for meromorphic functions}, Nihonkai Math. J., 24(2013), 121-134.
		%\bibitem{Ban_JMAA_2007} A. Banerjee, {\it On a question of Gross}, J. Math. Anal. Appl., 327(2)(2007), 1273-1283.
		%\bibitem{Ban_Port_2008} A. Banerjee, {\it Uniqueness of meromorphic functions sharing two sets with finite weight}, Port. Math. (N. S.), 65(1)(2008), 81-93. 
	%	\bibitem{Ban_Tamkang_2010} A. Banerjee, {\it Uniqueness of meromorphic functions sharing two sets with finite weight II}, Tamkang J. Math., 41(4)(2010), 379-392.
	%	\bibitem{Ban_Debrecen_2010} A. Banerjee, {\it Uniqueness of meromorphic function that share three sets II}, Publ. Math. Debrecen., 76(1-2)(2010), 173-182.
		%\bibitem{Ban_SABM_2010} A. Banerjee, {\it Some further results on the uniqueness of meromorphic functions sharing three sets}, Southeast Asian Bull. of Math., 34(2010), 835-846.
	%	\bibitem{Ban_Cubo_2011} A. Banerjee, {\it Some uniqueness results on meromorphic functions sharing three sets II}, CUBO A Math. J., 13(3)(2011), 57-67.
	%	\bibitem{Ban_Ban_Tbilisi_2023} A. Banerjee and J. Banerjee, {\it On strong uniqueness polynomials in the wider sense and associated unique range sets of the same category}, Adv. Stud: Euro-Tbilisi Math. J., 16(4)(2023), 175-189.
	%	\bibitem{Ban+bhattacharya_Slovaca_2011} A. Banerjee and P. Bhattacharjee, {\it Uniqueness and set sharing of derivatives of meromorphic functions},  Math. Slovaca., 61(2)(2011), 197-214.
	%	\bibitem{Ban+ I.L._cmft_2012} A. Banerjee and I. Lahiri, {\it A uniqueness polynomial generating a unique range set and vise versa}, Comput. Method Funct. Theo., 12(2)(2012), 527-539.
	%	\bibitem{Ban+Malk_CMFT_2017} A. Banerjee and S. Mallick, {\it On the characterisations of a new class of strong uniqueness polynomials generating unique range sets}, Comput. Methods Funct. Theo., 17(2017), 19-45.
	%	\bibitem{Ban+Mukherjee_hokkaido_2008} A. Banerjee and S. Mukherjee, {\it Uniqueness of meromorphic functions sharing two or three sets}, Hokkaido Math. J., 37(3)(2008), 507-530.
	%	\bibitem{Ban+Mukherjee_Sarajevo_2009} A. Banerjee and S. Mukherjee, {\it Some uniqueness results on meromorphic functions sharing two or three sets}, Sarajevo J. Math., 5(18)(2009), 169-189.
	
	%\bibitem{Chen+Yi_Resultsmath_2013}  Z. X. Chen, H. X. Yi, {\it On Sharing Values of Meromorphic Functions and Their Differences}, Results. Math., 63(2013), 557–565. 
%	\bibitem{Frank+Reinders_1998} G. Frank and M. Reinders, {\it A unique range set for meromorphic functions with 11 elements}, Com. Var. Theo. Appl., 37(1)(1998), 184-193.
%	\bibitem{Fujimoto_Amer.J.Math_2000} Fujimoto H., {\it On uniqueness of meromorphic functions sharing finite sets}, Amer. J. Math., 122(6)(2000), 1175-1203.
%	\bibitem{Fujimoto_Nagoya_2003} H. Fujimoto, {\it On uniqueness polynomials for meromorphic functions}, Nagoya Math. J., 170(2003), 33-46.
%	\bibitem{Gross_1976} F. Gross, {\it Factorization of meromorphic functions and some open problems}. Proc. Conf. Univ. Kentucky, Leixngton, Ky(1976); Lecture Notes in Math, 599(1977), 51-69, Springer(Berlin).
%\bibitem{Lahiri_Nagoya_2001}I. Lahiri, {\it Weighted sharing and uniqueness of meromorphic functions}, Nagoya Math. J., 161(2001), 193-206.
%\bibitem{Lahiri_IJMMS_2001} I. Lahiri,{\it Value distribution of certain differential polynomials}, Int. J. Math. Math. Sci., 28(2)(2001), 83-91.
%\bibitem{Lahiri_CVTA_2001}I. Lahiri, {\it Weighted value sharing and uniqueness of meromorphic functions}, Complex Var. Theo. Appl., 46(3)(2001), 241-253.
%\bibitem{Lin+Yi_Kyungpook_2003} W. C. Lin and H. X. Yi, {\it Some further results on meromorphic functions that share two sets}, Kyungpook Math. J., 43(1)(2003), 73-85.
%\bibitem{Lin+Yi_Com.Var_2003} W. C. Lin and H. X. Yi, {\it Uniqueness theorems for meromorphic functions that share three sets,} Complex Var. Theory Appl., 48(2003), 315-327.
%\bibitem{Lin+Yi_Kyungpook_2006} W. C. Lin and H. X. Yi, {\it Uniqueness of meromorphic functions and a question of Gross}, Kyungpook Math. J., 46(3)(2006), 437-444.
%\bibitem{Malk+Ban_Bohmica_2021} S. Mallick and A. Banerjee, {\it Uniqueness and two shared set problems of meromorphic functions in a different angle}, Math. Bohemica., 146(1)(2021), 19-45.
%\bibitem{Yi_1994} H. X. Yi, {\it Uniqueness of meromorphic functions and a question of gross}, J. Sc. China(Ser A), 37(7)(1994).
%\bibitem{Yi_Bullaus_1995} H. X. Yi, {\it Unicity theorems for meromorphic or entire functions II}, Bull. Austral. Math. Soc., 52(2)(1995), 215-224.
%\bibitem{Yi_Kodai_1999} H. X. Yi, {\it Meromorphic functions that share one or two values II}, Kodai Math. J., 22(2)(1999), 264-272. 
%\bibitem{Yi+Lu_Acta_2004} H. X. Yi and W. R. L$\ddot{u}$, {\it Meromorphic functions that share two sets II}, Acta Math. Sci. (Ser. B) Engl. Ed., 24(1)(2004), 83-90. 
%\bibitem{Wu+Lin_JCA_2015} A. D. Wu and W. C. Lin, {\it Some properties of meromorphic functions concerning shared-sets}, J. Classical Anal., 7(2015), 1-15.
	\end{thebibliography}

	\setcounter{page}{1}
	
	\title[Normal and differentiable periodicity of linear shift...... ]{\LARGE 
		N\huge\MakeLowercase {ormal and differentiable periodicity of linear shift operators under partial sharing}}
	
	\author[A. Banerjee, J. Banerjee]{Abhijit Banerjee $^1$, Jhilik Banerjee$^{*}$ $^2$ }
	
	\address{$^{1}$ Department of Mathematics, University of Kalyani, West Bengal 741235, India.}
	\email{abanerjee\_kal@yahoo.co.in}
	\address{$^{2}$ Department of Mathematics, University of Kalyani, West Bengal 741235, India.}
	\email{ jhilikbanerjee38@gmail.com}
	
	
	%\dedicatory{This paper is dedicated to Professor ABCD}
	
	\subjclass[2010]{30D35.}
	
	\keywords{Meromorphic functions; partially CM sharing;  partially IM sharing; linear shift operator, order.}
	
	\date{Received: xxxxxx; Revised: yyyyyy; Accepted: zzzzzz.
		\newline \indent $^{*}$ Corresponding author}
	
	\begin{abstract}
	
	In this paper, we investigate the periodicity of linear shift operators associated with meromorphic functions under the framework of partial sharing conditions. Furthermore, we introduce the novel concept of differentiable periodicity, which provides a robust framework for analyzing the behavior of these operators under partial CM or IM sharing. By establishing comprehensive criteria for differentiable periodicity, this work offers a new perspective in the theory of shift operators. The findings not only extend but also refine and improve prior results, such as those presented in the papers of W J. Chen, Z. G. Huang (2022) and D. C. Pramanik, A. Sarkar (2024).
	
		%The primary objective of this paper is to determine sufficient conditions under which a linear shift operator exhibits periodicity. The investigation is based on the partial value-sharing criterion of meromorphic functions, with multiplicities disregarded. Notably, this work represents the first comprehensive result establishing the periodicity of a linear shift operator in its entirety. The findings presented in this study significantly extend and refine earlier results in the field.     

	  
	\end{abstract} \maketitle
	
	\section{Introduction and background}
At the outset, we introduce the following key notations, which are essential for establishing a comprehensive framework to analyze meromorphic functions and their behavior concerning shared values or sets. 
	\par Let  $f(z)$ be a non-constant meromorphic function and $a \in \mathbb{C}$. Define the function $\nu_{f}^{a}: \mathbb{C}\rightarrow \mathbb{N}$ by 
	\[\nu^{a}_f(z)=\begin{cases}
		0 &\text{if }f(z) \not =a\\d & \text{if }f(z)=a \text{ with multiplicity d}, 
	\end{cases}\] and set $\nu^{\infty}_{f}(z)=\nu^{0}_{\frac{1}{f}}(z)$. Next, define the reduced counting function $\ol \nu_{f}^{a} : \mathbb{C} \rightarrow \mathbb{N}$ as: $$\ol \nu_{f}^{a}(z) =\text{min} \{ \nu_{f}^{a}(z) ,1\}$$ and similarly $\ol \nu^{\infty}_{f}(z)=\ol \nu^{0}_{\frac{1}{f}}(z)$. Next for $m \in \mathbb{N}$ and $a\in \mathbb{C} \cup \{\infty\}$, define the reduced multiplicity counting function $\nu^{\leq m}_{(f,a)}:\mathbb{C} \cup \{\infty\} \rightarrow \mathbb{N} $ by 
\[\nu^{\leq m}_{(f,a)}(z)=\begin{cases}
	0 &\text{if }\nu^{a}_f(z) > m\\\nu^{a}_f & \text{if }\nu^{a}_f(z)\leq m
\end{cases}\]  and $\ol \nu^{\leq m}_{(f,a)}(z)=\text{min}\left\{\nu^{\leq m}_{(f,a)}(z),1\right\}$. When $m=\infty$, we interpret $\nu^{\leq m}_{(f,a)}(z)$ as $\nu^{a}_f(z) $.\par For $S \subset \mathbb{C}\cup \{\infty\}$, define the set $E(S,f)$  as \beas E(S,f) = \bigcup_{a \in S}\{(z,\nu^{a}_f(z)): z \in \mathbb{C}\}.\eeas \par In an analogous way, replacing $\nu^{a}_f(z)$ with $\ol \nu^{a}_f(z)$ in $ E(S,f)$, define $\ol E(S,f)$ as: 
\beas \ol E(S,f) = \bigcup_{a \in S}\{(z,\ol\nu^{a}_f(z)): z \in \mathbb{C}\}.\eeas
Using these definitions, we describe various types of set sharing between meromorphic functions $f$ and $g$:\\ $\bullet$ If $E(S,f)=E(S,g)$,  we say that $f$ and $g$ share the set $S$ CM.\\ $\bullet$  If $\ol E(S,f)=\ol E(S,g)$,  we say that $f$ and $g$ share the set $S$ IM.
\\$\bullet$ If $E(S,f) \subseteq E(S,g)$,  we say that $f$ and $g$ partially share $S$ CM. \\$\bullet$ If $\ol E(S,f) \subseteq \ol E(S,g)$,  we say that $f$ and $g$ partially share $S$ IM.\\
When $S$ is a singleton, the above definition corresponds to value sharing.
			%For a non-constant meromorphic function $f(z)$ and $S \subset \mathbb{C}\cup \{\infty\}$, let $E_f (S) = \bigcup_{a\in S}\{(z,p) \in \mathbb{C}\times\mathbb{N}: f(z)=a \text{ with multiplicity $p$}\} $ ($\ol E_f(S)=\bigcup_{a\in S}\{(z,1) \in \mathbb{C}\times\mathbb{N}: f(z)=a$\}.
	%	If $E(S,f)=E(S,g)\; (\ol E(S,f)=\ol E(S,g))$ we say that $f$ and $g$ share the set $S$ CM (IM). When $S$ is a singleton set, the definition coincides with the traditional definition of CM (IM) value sharing. In this context, if $E(S,f) \subseteq E(S,g)\;( \ol E(S,f) \subseteq \ol E(S,g))$, we say that $f(z)$ and $g(z)$ partially share $S$ CM (IM) i.e., counting multiplicities (ignoring multiplicities). 
		%If $\ol E(S,f) \subseteq \ol E(S,g)$, we say that $f(z)$ and $g(z)$ partially shares $aS$ IM (ignoring multiplicities). 
	%	If $S=\{a\}$ i.e., is a singleton set, then we say $f$ and $g$ share a value $a$ CM (IM).
			%	At the outset, we assume that the readers are acquainted with the conventional notations of value distribution theory defined on whole complex plane $\mathbb{C}$ such as $N(r,f)$, $m(r,f)$ $T(r,f)$, $S(r,f)$, etc, as outlined in \cite{Hayman_1964}, \cite{Yang_Springer_1993}. So we refrain from providing detail explanations. \par In value distribution theory we generally concern about the distribution of the zeros of the function $f(z)-a$, where $a\in \mathbb{C}$ and form which the idea of sharing of values or sets comes under consideration. For the standard notations of set sharing, we refer the readers to make a glance over the relevant information provided in the second paragraph of \cite{Ban_Georgian_2008}, which automatically includes the definition of value sharing. Below we invoke it.\par 
		
		
		%Let $S$ be a set of distinct elements of $\mathbb{C}\cup\{\infty\}$. We denote by $E_{f}(S)=E(S,f)=\everymath{\displaystyle}\bigcup_{a\in S}\{z: f(z)-a=0\}$, where each zero is counted according to its multiplicity. If multiplicities are not taken into account, the set is denoted by $\ol E(S,f)$.
%	Let $a \in \mathbb{C}\cup\{\infty\}$. We denote by $E(a,f)=\{z: f(z)-a=0\}$, where each zero is counted according to its multiplicity. If multiplicities are not taken into account, the set is denoted by $\ol E(a,f)=\{z: f(z)-a=0\}$.  In this context, if $E(a,f) \subseteq E(a,g)$, we say that $f(z)$ and $g(z)$ partially share $a$ CM (counting multiplicities). If $\ol E(a,f) \subseteq \ol E(a,g)$, we say that $f(z)$ and $g(z)$ partially shares $a$ IM (ignoring multiplicities).
	\par For further information or a detailed explanation of these concepts, readers are encouraged to consult the original sources cited in the manuscript: \cite{Ban_Georgian_2008}. Additionally, we assume that readers are familiar with the standard notations of value distribution theory on the entire complex plane $\mathbb{C}$, including $N(r,f)$, $m(r,f)$, $T(r,f)$, $S(r,f)$ etc. as outlined in \cite{Hayman_1964} and \cite{Yang_Springer_1993}.
	% Now, we shall calculate the counting functions of zeros of $L_{j}f^{(l)}(z)-a$ with multiplicity of $\geq 2,$ which is denoted by $N_{2)} \left(r, \frac{1}{L_{j}f^{(l)}(z)-a}\right)$ counted according to its multiplicities.
	 We use the notation $N_{(k}(r,a,f(z))$ to denote the counting function of $a$-points of $f(z)$ of multiplicity $\geq k$.
	For $a\in \mathbb{C}\cup\infty$, we also use the notation $\ol N_{*}(r,a;f,g)$ to denote the reduced counting function of common $a$-points of $f$ and $g$ of different multiplicities.
	 \par The order and  hyper-order of $f$ are defined respectively as follows:\beas \rho(f)=  \limsup_{r\longrightarrow \infty} \frac{\log T(r,f)}{\log r}, \;\;\;\;\; \rho_2(f)=\limsup_{r\longrightarrow \infty} \frac{\log \log T(r,f)}{\log r}.\eeas 
	\par Further throughout the paper, we will assume $c \in \mathbb{C} \setminus\{0\}$ and the shift operator of $f$ will be denoted by $f(z+c)$. \par
	We begin our discussion with a result established by Qi  \cite{Qi_Ann polon_2011}.
	\begin{theoA}\cite{Qi_Ann polon_2011}
		Let $f$ be a entire function of finite order, $c\in \mathbb{C}$ and $a_i$, $i=1,2$ be three distinct periodic functions with period $c$. If $f(z)$ and $f(z+c)$ share $a_1$, $a_2$ IM, then $f(z)=f(z+c)$.
	\end{theoA} 
	
	For meromorphic function some analogous results are as follows: 
	\begin{theoB}\cite{Heittokangas_Com. var_2011}
		Let $f$ be a meromorphic function of finite order and $c \in \mathbb{C}$. If $f(z)$ and $f(z+c)$ share three distinct periodic functions $a_i$, $i=1,2,3$ with period $c$ CM, then $f(z)=f(z+c)$.
	\end{theoB}

\begin{theoC}\cite{Heittokangas_JMAA_2009}
 Under the same condition of {\em Theorem B}, if $f(z)$ and $f(z+c)$ share $a_1$, $a_2$ CM and $a_3$ IM, then $f(z)=f(z+c)$.
\end{theoC}
In this respect, further continuations of research can be found in \cite{Chen+Xu_Houston_2017} and \cite{Li+Yi_Bull.Korean_2016}.
%\begin{theoD}
%	Let $f$ be a non-constant meromorphic function of $\rho_2(f)<1$ and let $a_1,a_2,a_3 \in \mathbb{C} \cup \{\infty\}$. If $f(z)$ and $f(z+c)$ share $a_1$, $a_2$ CM and $a_3$ IM, where $c \in \mathbb{C}\setminus\{0\}$, then $f(z) \equiv f(z+c)$.
%\end{theoD}
%\begin{theoE}
%	Let $f$ be a non-constant entire function of $\rho_2(f)<1$ and $a_1$, $a_2$ be two distinct finite values. If $f(z)$ and $f(z+c)$ share $a_1$, $a_2$ IM, where $c  \in \mathbb{C}\setminus\{0\} $,then $f(z) \equiv f(z+c).$
%\end{theoE}
 Recently, Chen et al. introduced new types of structures that had not been considered earlier. In their work, Chen and Huang \cite{Chen+Huang_Contemp_2022} presented results focusing on partial and normal sharing, as outlined below:
\begin{theoD}\cite{Chen+Huang_Contemp_2022}
	Let $f(z)$ be a meromorphic function of finite order, $a \in \mathbb{C}\setminus \{0\}$ and $l,s$ be integers with $0 \leq l<s$. If $f^{(l)}(z)$ and $f^{(s)}(z+c)$ share $a$ CM and satisfy $E(0,f^{(l)}(z))  \subset E(0, f^{(s)}(z+c) )$ and $E(\infty, f^{(s)}(z+c)) \subset E(\infty, f^{(l)}(z) )$, then $f^{(l)}(z) \equiv f^{(s)}(z+c) $.
\end{theoD}
\begin{theoE}\cite{Chen+Huang_Contemp_2022}
	Let $f(z)$ be a transcendental meromorphic function of finite order, $0 \leq l<s$. Suppose that, $f^{(l)}(z)$ and $f^{(s)}(z+c)$ share a finite value $a(\not = 0)$ IM and satisfy $E(0,f^{(l)}(z)) \subset E(0,f^{(s)}(z+c))$ and $E(\infty, f^{(s)}(z+c)) \subset E(\infty, f^{(l)}(z))$. If $N\left(r, \frac{1}{f^{(l)}(z)}\right)+\ol N\left(r, \frac{1}{f(z)}\right)=S(r,f)$, then $f^{(l)}(z)\equiv f^{(s)}(z+c) $.
\end{theoE} 
Very recently, in \cite{Pramanik+Sarkar_Math.Studii_2024} the authors have been able to successfully remove the conditions over counting function in Theorem E. The result is as follows:
\begin{theoF}\cite{Pramanik+Sarkar_Math.Studii_2024}
	Let $f(z)$ be a non-constant meromorphic function of finite order, $0\leq l<s$. If $f^{(l)}(z)$ and $f^{(s)}(z+c)$ share a finite value $a(\not = 0)$ IM and satisfy $E(0,f^{(l)}(z)) \subset E(0,f^{(s)}(z+c))$ and $E(\infty, f^{(s)}(z+c)) \subset E(\infty, f^{(l)}(z))$, then $f^{(l)}(z)\equiv f^{(s)}(z+c) $.
\end{theoF}

\section{Motivation, main results and relevant examples}

The following example shows that Theorem A can not be extended for meromorphic function under partial sharing.
\begin{exm}
	Let us assume two distinct complex numbers $a,b$, $c=\pi i$ and $f(z)=\frac{ae^z-b}{e^z-1}$. Then, $f(z+c)=\frac{ae^z+b}{e^z+1}$. It is easy to check, $ E(a,f(z)) = E(a, f(z+c))$ and $ E(b,f(z)) =  E(b, f(z+c))$, but $f(z) \not = f(z+c).$
\end{exm} 
We see that in the above example the poles are not shared. 
In light of the above example, an intriguing question arises:

\par Can the periodicity of a meromorphic function be achieved under partial sharing on $\mathbb{C}$? If so, what would be the nature of the pole sharing?

The purpose of this paper is to address these questions and contribute to this line of inquiry. Specifically, we aim to establish our results within a more generalized framework, namely the linear shift operator, which is defined as follows:

For a non-constant meromorphic function $f(z)$, a non negative integer $j$, $a_i \in \mathbb{C}$, $a_j\not=  0$ and $c \in \mathbb{C}$, define $ L_{j}f(z)$ as
\bea\label{e1.1} L_{j}f(z)=\sum_{i=0}^{j} a_{i}f(z+ic).\eea


The following theorem is one of our main results, where we investigate the periodicity of the linear shift operator based solely on partial IM sharing which was not addressed earlier. 
\begin{theo}\label{t1}
	Let $f(z)$ be a non-constant meromorphic functions with $\rho_2(f) <1$ and let $a$, $b \in \mathbb{C} $ be two distinct numbers. Let $\ol E(a,L_{j}f(z)) \subseteq \ol E(a, L_{j}f(z+c)) $, $\ol E(b,L_{j}f(z)) \subseteq \ol E(b, L_{j}f(z+c)) $, $E(\infty,L_{j}f(z+c)) \subseteq E(\infty, L_{j}f(z))$ with $\ol N(r,f) =S(r,f)$,  then $L_{j}f(z)=L_{j}f(z+c)$.
\end{theo} 
The following example shows that, by no means, the linear shift operator $L_{j}(z+c)$ in Theorem \ref{t1}, can not be replaced by an arbitrary linear shift operator.
\begin{exm}
	Consider the function $L_{0}f(z)=f(z)=\frac{e^{\frac{2\pi i z}{c}}-1}{e^{\frac{2\pi i z}{c}}+1},$ where $c \in \mathbb{C}\setminus\{0\}$. Choose a linear shift operator $\tilde{L}_{3}f(z)$ by \beas\tilde{L}_{3}f(z)&=&3f(z+3c)-f(z+2c)-2f(z+c)-f(z)\\&=& -\frac{e^{\frac{2\pi i z}{c}}-1}{e^{2\frac{\pi i z}{c}}+1}. \eeas Clearly, $E(1,L_{0}f(z))=E(1,\tilde{L}_{3}f(z))$, $E(0,L_{0}f(z))=E(0,\tilde{L}_{3}f(z))$ and $E(\infty,L_{0}f(z))\\=E(\infty,\tilde{L}_{3}f(z))$, but $f(z) \not = \tilde{L}_{3}f(z).$
\end{exm}
The next example shows that for entire functions, in Theorem \ref{t1}, $f(z+c)$ can not be replaced by $\Delta_{c}f(z)=f(z+c)-f(z)$.
\begin{exm}
	Take $L_{0}f(z)=f(z)=\left(\frac{e^{\frac{\pi i z}{c}}-1}{2}\right)^2+1$. Then $\Delta_{c}f(z)= e^{\frac{\pi i z}{c}}$. Here, $f$ and $\Delta_{c}f$ share $1$, $0$ IM, but $f \not = \Delta_{c}f$.
\end{exm}
Next, we define differentiable periodic function.
\begin{defi}
	Let $F(z)$ be a non-constant meromorphic function and $l$, $s$ two non-negative integers. $F(z)$ is said to be differentiably periodic if $F^{(l)}(z) \equiv F^{(s)}(z+c)$, where $0  \leq l <s$.\end{defi}

In light of Theorem D, it becomes of significant interest to investigate the differentiable periodic property of $L_{j}f(z)$ as defined in (\ref{e1.1}) as this perspective provides a deeper understanding of the behavior of the linear shift operator $L_{j}$ under certain sharing conditions. Moreover, by relaxing the condition from CM sharing of the non-zero value $a$ to partial CM sharing of $a$, it can be shown that a similar result to Theorem D holds. The following theorem formalizes this extension:
%In this context, we present the following theorem, which demonstrates that the sharing conditions outlined in Theorem \ref{e1.1} can be relaxed in expense of allowing partial CM sharing to partial IM sharing, thereby broadening the scope of the result.
\begin{theo}\label{t2}
	Let $f(z)$ be a non-constant meromorphic function with $\rho_2(f)<1$, $a \in \mathbb{C}\setminus\{0\}$ and $l$, $s$ be two integers such that $0\leq l<s$. If $ E(a,L_{j}f^{(l)}(z)) \subseteq  E(a,  L_{j}f^{(s)}(z+c))$, $E(0,L_{j}f^{(l)}(z)) \subseteq  E(0,  L_{j}f^{(s)}(z+c))$, $  E(\infty,  L_{j}f^{(s)}(z+c)) \subseteq E(\infty,L_{j}f^{(l)}(z)) $, then $L_{j}f^{(l)}(z) \equiv L_{j}f^{(s)}(z+c). $
\end{theo}
\begin{note}
	We can see that {\em Theorem \ref{t2}} is not only extending {\em Theorem D}, but also improving it by relaxing sharing condition.
\end{note}
The following example shows that, sharing of the non-zero element in Theorem \ref{t2} can not be dropped.
\begin{exm}
	Set $L_{0}f(z)=\sin z$, then $L_{0}f^{\prime}(z+\frac{\pi}{2})=-\sin z$. Clearly, $E(0, L_{0}f(z))=E(0,L_{0}f^{\prime}(z+\frac{\pi}{2}))$ and $E(\infty, L_{0}f(z))=E(\infty,L_{0}f^{\prime}(z+\frac{\pi}{2}))$, but $E(1, L_{0}f(z))\not \subset E(1,L_{0}f^{\prime}(z+\frac{\pi}{2}))$ and $L_{0}f(z) \not = L_{0}f^{\prime}(z+\frac{\pi}{2}).$
\end{exm}

In view of Theorem \ref{t1}, it is timely and significant to investigate the differentiable periodic property of $L_{j}f(z)$ under the framework of partial IM value sharing over $\mathbb{C}$.
\begin{theo}\label{t3}
Under the same situation of {\em Theorem \ref{t2}}, if $\ol E(a,L_{j}f^{(l)}(z)) \subseteq \ol E(a,  L_{j}f^{(s)}(z+c))$, $E(0,L_{j}f^{(l)}(z)) \subseteq  E(0,  L_{j}f^{(s)}(z+c))$ and $  E(\infty,  L_{j}f^{(s)}(z+c)) \subseteq E(\infty,L_{j}f^{(l)}(z)) $ and $\ol N\left(r, \frac{1}{L_{j}f^{(l)}(z)}\right)=S(r,f)$, then $L_{j}f^{(l)}(z) \equiv L_{j}f^{(s)}(z+c). $
\end{theo}
The following example shows that sharing structure of $\infty$ in Theorem \ref{t3} can not be reversed.
\begin{exm}
	Take $L_{0}f(z)=\frac{2e^{\frac{2\pi i z}{c}}}{e^{\frac{2\pi i z}{c}}-1},$ where $c \in \mathbb{C}\setminus\{0\}$. Then, $L_{0}f^{\prime}(z+c)=\frac{-4e^{\frac{2\pi i z}{c}}}{(e^{\frac{2\pi i z}{c}}-1)^2}$. Clearly, $\ol E(1, L_{0}f(z)) \subset\ol E(1,L_{0}f^{\prime}(z+c))$, $E(0, L_{0}f(z)) =E(0,L_{0}f^{\prime}(z+c))$ and $\ol N\left(r,\frac{1}{L_{0}f(z)}\right)=S(r,f)$, but $E(\infty, L_{0}f^{\prime}(z+c)) \not \subset E(\infty,\\ L_{0}f(z))$ and $L_{0}f(z) \not = L_{0}f^{\prime}(z+c)$.
	
\end{exm}
The next theorem extends and refines Theorems E and F to the case of differentiably periodic linear shift operators. In fact, to prove the theorem, we have adopted a different approach than the one used in \cite{Chen+Huang_Contemp_2022} and \cite{Pramanik+Sarkar_Math.Studii_2024}. 
\begin{theo}\label{t4}
	Under the same situation of {\em Theorem \ref{t2}}, if $\ol E(a,L_{j}f^{(l)}(z)) = \ol E(a,  L_{j}f^{(s)}(z+c))$, $E(0,L_{j}f^{(l)}(z)) \subseteq  E(0,  L_{j}f^{(s)}(z+c))$ and $  E(\infty,  L_{j}f^{(s)}(z+c)) \subseteq E(\infty,L_{j}f^{(l)}(z)) $, then $L_{j}f^{(l)}(z) \equiv L_{j}f^{(s)}(z+c). $
\end{theo}
The following two examples show that Theorem \ref{t1} cannot be directly extended to set sharing structure.

\begin{exm}
	Let $L_{0}f(z)= \frac{2e^{\frac{\pi i z}{c}}}{1+e^{\frac{2\pi i z}{c}}},$ $c \in \mathbb{C}\setminus \{0\}$. Then we see that for a non-zero constant $\alpha$, $E(\{\alpha,-\alpha\},L_{0}f(z))=E(\{\alpha,-\alpha\},L_{0}f(z+c))$, $E(\infty,L_{0}f(z))=E(\infty,L_{0}f(z+c))$
	 %$L_{0}f(z)$ and $L_{0}f(z+c)$ share $\{\alpha, -\alpha\} $ CM,
	  but $L_{0}f(z) \not = L_{0}f(z+c)$.
\end{exm}
\begin{exm}
Let $L_{0}f(z)= \frac{1+e^{\frac{\pi i z}{c}}}{1-e^{\frac{2\pi i z}{c}}},$ $c \in \mathbb{C}\setminus \{0\}$. Then we see that for a non-zero constant $\alpha$, $E\left(\{\alpha,\frac{1}{\alpha}\},L_{0}f(z)\right)=E\left(\{\alpha,\frac{1}{\alpha}\},L_{0}f(z+c)\right)$, $E(\infty,L_{0}f(z))=E(\infty,L_{0}f(z+c))$
%$L_{0}f(z)$ and $L_{0}f(z+c)$ share $\{\alpha, \frac{1}{\alpha}\} $ CM,
 but $L_{0}f(z) \not = L_{0}f(z+c)$.
\end{exm}
 In view of Example 2.6 and 2.7, the following question is inevitable:
\begin{ques}
	Can we can replace the value sharing condition with set sharing condition in the {\em Theorems \ref{t1}-\ref{t4}}?
\end{ques}
Below we are presenting the following chart, which provides an in-depth comparison of the related theorems in terms of sufficient conditions in a compact form. Throughout the chart, we assume that  $a(\not = 0)$, $b$ are two distinct complex numbers.  

\begin{center}
	\includegraphics[scale=.51]{Table_3}
\end{center}
\section{Lemmas}
In this section, we are going to present some lemmas, which will be needed to proceed further.
	\begin{lem}\label{l3.1}\cite{Halburd+Korhonen_Tohge_TAMS_2014}
		Let $f(z)$ be a meromorphic function with $\rho_2(f)<1$ and $c\in \mathbb{C}$. Then we have \beas m\left(r, \frac{f(z+c)}{f(z)}\right)+m\left(r \frac{f(z)}{f(z+c)}\right)=S(r,f), \eeas where $S(r,f)=o(T(r,f))$, for all $r$, outside of a possible exceptional set $S$ with finite linear measure. 
	\end{lem}
	\begin{lem}\label{l3.2}\cite{Chiang+Feng_Ramanujan_2008}
		Let $f(z)$ be a non-constant meromorphic function of $\rho_2(f)<1$ and $c \in \mathbb{C}$. Then {\small \beas N(r,f(z+c))=N(r,f)+S(r,f),\; N\left(r, \frac{1}{f(z+c)}\right)= N\left(r,\frac{1}{f(z)}\right)+S(r,f)\eeas} and {\small \beas \ol{N}(r,f(z+c))=\ol{N}(r,f)+S(r,f),\; \ol{N}\left(r, \frac{1}{f(z+c)}\right)= \ol{N}\left(r,\frac{1}{f(z)}\right)+S(r,f).\eeas} 
	\end{lem}
\begin{lem}\label{l3.3}\cite{Yi+Yang_2003_Sci.Press}
	Let $f(z)$ be a non-constant meromorphic function and let $d_j, (j = 1,2,\ldots, q)$
	be $q$ distinct complex numbers. Then we have \beas m\left(r,\sum_{i=1}^{q}\frac{1}{f-d_i}\right)=\sum_{i=1}^{q}m\left(r,\frac{1}{f-d_i}\right)+O(1). \eeas
\end{lem}
\begin{lem}\label{l3.3e}\cite{Ban+Roy_Contemp_2023}
Let $f(z)$ be a meromorphic function of $\rho_2(f) < 1$, then we have
\beas  T(r, f(z+c)) = T(r, f(z)) + S(r, f).\eeas
\end{lem}
%\begin{lem}\label{l3.3e}\cite{Halburd+Korhonen_Tohge_TAMS_2014}
% Let $T : [0,+\infty) \rightarrow [0,+\infty)$ be a non-decreasing continuous function and let $ s \in(0,+\infty) $. If $\rho_2(T)<1$ and $\delta \in (0,1-\rho_2)$, then \beas T(r+s)=T(r)+o\left(\frac{T(r)}{r^\delta}\right) ,\eeas  where $r$ runs to infinity outside of a set of finite logarithmic measures.
%\end{lem}
%\begin{lem}\label{l3.3}\cite{Chiang+Feng_Ramanujan_2008}
%Let $f(z)$ be a meromorphic function of finite order $\rho$ and $c$ be a non-zero complex constant. Then for each $\epsilon >0$, we have \beas T(r,f(z+c))=T(r,f(z))+O(r^{\rho-1+\epsilon})+O(\log r).\eeas
%\end{lem}
\begin{lem}\label{l3.4}
	Let $f(z)$ be a meromorphic function of $\rho_2(f)<1$, then $S(r, L_jf(z))$ and $S(r, L_jf(z+c))$ can be replaced by $S(r,f)$.
\end{lem}
\begin{proof}
	In view of {\it Lemmas \ref{l3.1}, \ref{l3.3e}}, we have {\small \beas T(r, L_{j}f(z))&=&m(r,L_{j}f(z))+N(r,L_{j}f(z))\\&\leq& m\left(r,\frac{L_{j}f(z)}{f}\right)+m(r,f(z))+N(r,L_{j}f(z))+S(r,f) \\&\leq& (j+2)T(r,f(z))+S(r,f)\eeas} and so the lemma follows. 
	Similar result holds for $S(r, L_jf(z+c))$.
	% $S(r,L_{j}f^{(l)}(z))$ and $S(r,L_{j}f^{(s)}(z+c))$. 
	\end{proof}
\begin{lem}\label{l3.4e}
	Let $f(z)$ be a non-constant meromorphic function with $\rho_2(f)<1$, $c \in \mathbb{C}\setminus\{0\}$ and $l$, $s \geq 0$. Then, $S(r,L_{j}f^{(l)}(z))$ and $S(r,L_{j}f^{(s)}(z+c))$ can be replaced by $S(r,f)$.
\end{lem}
\begin{proof}
	In view of {\it Lemmas \ref{l3.1}. \ref{l3.3e}} and logarithmic derivative theorem, we obtain
	{\small\beas T(r, L_{j}f^{(l)}(z))&=&m(r,L_{j}f^{(l)}(z))+N(r,L_{j}f^{(l)}(z))\\&\leq& m\left(r,\frac{L_{j}f^{(l)}(z)}{L_{j}f(z)}\right)+m\left(r,\frac{L_{j}f(z)}{f(z)}\right)+N(r,L_{j}f^{(l)}(z))+S(r,f)\\&\leq& (l+j+1)T(r,f(z))+S(r,f). \eeas} Hence, the lemma follows. Similar results holds for  $S(r,L_{j}f^{(s)}(z+c))$.
\end{proof}
\begin{lem}\label{l3.5}
	Let $f(z)$ be a non-constant meromorphic function  with $\rho_2(f)<1$. For any complex number $b$ we have, \beas m\left(r, \frac{ L_{j}f^{(s)}(z+c)}{L_{j}f^{(l)}(z)-b}\right)\leq S(r,f),\;\;\text{where } 0 \leq l<s.\eeas 
\end{lem}
\begin{proof}
	Using logarithmic derivative theorem and {\it Lemmas \ref{l3.1}, \ref{l3.4e}}, we obtain \beas  m\left(r, \frac{ L_{j}f^{(s)}(z+c)}{L_{j}f^{(l)}(z)-b}\right) &\leq &m\left(r, \frac{ L_{j}f^{(s)}(z+c)}{L_{j}f^{(s)}(z)}\right)+m\left(r, \frac{L_{j}f^{(s)}(z)}{L_{j}f^{(l)}(z)-b}\right)  \\&=& S(r,f). \eeas
\end{proof}
\begin{lem}\label{l3.6}
	Under the same situation of {\em Lemma \ref{l3.5}}, if $a \in \mathbb{C}\setminus\{0\}$ and $0\leq l< s$; then \beas &&m\left(r,\frac{1}{L_{j}f^{(l)}(z)}\right)+m\left(r, \frac{1}{L_{j}f^{(l)}(z)-a}\right) \\&&\leq m(r,L_{j}f^{(l)}(z))+N(r, L_{j}f^{(s)}(z+c) )-N\left(r, \frac{1}{ L_{j}f^{(s)}(z+c)}\right)+S(r,f). \eeas
\end{lem}

\begin{proof}
	Set \beas F(z)=\frac{1}{L_{j}f^{(l)}(z)}+\frac{1}{L_{j}f^{(l)}(z)-a}.\eeas Further, using {\it Lemmas \ref{l3.4e}, \ref{l3.5}} we get \beas m(r, L_{j}f^{(s)}(z+c)F(z)) &\leq& m\left(r,\frac{ L_{j}f^{(s)}(z+c)}{L_{j}f^{(l)}(z)}\right)+m\left(r,\frac{ L_{j}f^{(s)}(z+c)}{L_{j}f^{(l)}(z)-a}\right)\\&\leq& S(r,f).\eeas Thus, \beas m(r, F(z)) &\leq& m(r, L_{j}f^{(s)}(z+c)F(z))+m\left(r, \frac{1}{ L_{j}f^{(s)}(z+c)}\right) \\&\leq&m\left(r, \frac{1}{ L_{j}f^{(s)}(z+c)}\right)+S(r,f).\eeas Next, using {\it Lemma \ref{l3.3}}, the First Fundamental Theorem and the above inequality we obtain, \beas  T(r,  L_{j}f^{(s)}(z+c))&=&T\left(r, \frac{1}{ L_{j}f^{(s)}(z+c)}\right)+O(1)\\&=&N\left(r, \frac{1}{ L_{j}f^{(s)}(z+c)}\right)+m\left(r, \frac{1}{ L_{j}f^{(s)}(z+c)}\right)+S(r,f)\\&\geq &  m(r,F(z))+N\left(r, \frac{1}{ L_{j}f^{(s)}(z+c)}\right)+S(r,f)\\&=&m\left(r,\frac{1}{L_{j}f^{(l)}(z)}\right)+m\left(r,\frac{1}{L_{j}f^{(l)}(z)-a}\right)\\&&+N\left(r, \frac{1}{ L_{j}f^{(s)}(z+c)}\right)+S(r,f).\eeas Thus, from the above inequality using {\it Lemmas \ref{l3.1}, \ref{l3.4e}, \ref{l3.5}} we have \beas&& m\left(r,\frac{1}{L_{j}f^{(l)}(z)}\right)+m\left(r,\frac{1}{L_{j}f^{(l)}(z)-a}\right) \\&\leq&  T(r,  L_{j}f^{(s)}(z+c))-N\left(r, \frac{1}{ L_{j}f^{(s)}(z+c)}\right)+S(r,f)\\&=&m(r,  L_{j}f^{(s)}(z+c))+N(r,  L_{j}f^{(s)}(z+c))-N\left(r, \frac{1}{ L_{j}f^{(s)}(z+c)}\right)+S(r,f)\\&\leq&m\left(r,\frac{ L_{j}f^{(s)}(z+c)}{L_{j}f^{(l)}(z)}\right)+m(r,L_{j}f^{(l)}(z))+N(r,  L_{j}f^{(s)}(z+c))\\&&-N\left(r, \frac{1}{ L_{j}f^{(s)}(z+c)}\right)+S(r,f)\\&\leq & m(r,L_{j}f^{(l)}(z))+N(r,  L_{j}f^{(s)}(z+c))-N\left(r, \frac{1}{ L_{j}f^{(s)}(z+c)}\right)+S(r,f).\eeas
\end{proof}
\begin{lem}\label{l3.9}
	Let $f(z)$ a meromorphic function of $\rho_2(f)<1$ and $a \in \mathbb{C}\setminus\{0\}$. If $  E(\infty,  L_{j}f^{(s)}(z+c)) \subseteq E(\infty,L_{j}f^{(l)}(z)) $ for $0\leq l<s$, then $\ol N(r,L_{j}f^{(l)}(z))=\ol N(r,L_{j}f^{(s)}(z+c))=S(r,f)$.
\end{lem}
\begin{proof}
	Since $  E(\infty,  L_{j}f^{(s)}(z+c)) \subseteq E(\infty,L_{j}f^{(l)}(z)) $, we deduce \beas N(r,L_{j}f^{(s)}(z+c)) \leq N(r,L_{j}f^{(l)}(z))+S(r,f). \eeas Using {\it Lemma \ref{l3.4e}} we get, \beas  N(r,L_{j}f^{(s)}(z+c))&=&N(r,L_{j}f^{(s)}(z))+S(r,f)\\&=&N(r,L_{j}f^{(l)}(z))+(s-l)\ol N(r,L_{j}f^{(l)}(z))+S(r,f)\\&\leq & N(r,L_{j}f^{(l)}(z))+S(r,f),\eeas i.e., $\ol N(r,L_{j}f^{(l)}(z))=\ol N(r,L_{j}f^{(s)}(z+c))=S(r,f)$.
\end{proof}
%\begin{lem}\label{l3.10}
%	Let $f(z)$ be a non-constant meromorphic function of $\rho_2(f)<1$. If $E(0,L_{j}f^{(l)}(z)) \subseteq  E(0,  L_{j}f^{(s)}(z+c))$ and $  E(\infty,  L_{j}f^{(s)}(z+c)) \subseteq E(\infty,L_{j}f^{(l)}(z)) $, then $L_jf(z)$ is transcendental.
%\end{lem}
%\begin{proof}
%	Clearly, $\rho_2(L_jf(z)) \leq \rho_2(f) <1$. If $\rho_2(L_jf(z))$ is of infinite order, then $L_jf(z)$ can not be a rational function. If $\rho_2(L_jf(z))$ is finite then  
%	proceeding as the same way of {\it Lemma 4 of \cite{Pramanik+Sarkar_Math.Studii_2024}} we can prove that $L_jf(z)$ is transcendental.
%\end{proof}
\section{Proofs of the Theorems}
	\begin{proof}[\underline{\em\bf{Proof of Theorem \ref{t1}}}]The proof of the theorem is inspired by the idea of Lin-Ishizaki \cite{Lin+Ishizaki_JMAA_2018}.\\ 
On the contrary, we assume that, $L_{j}f(z) \not \equiv L_{j}f(z+c)$. Take a function \bea\label{e4.1} \Phi(z)= \frac{L_{j}f^\prime(z)(L_{j}f(z)-L_{j}f(z+c))}{(L_{j}f(z)-a)(L_{j}f(z)-b)}. \eea
$\bullet$ Let $z_0$ be an $a$- point of $L_{j}f(z)$ of order $p$ and same of $L_{j}f(z+c)$ of order $q$, then $z_0$ is a zero of $\Phi(z)$ of order $(p-1)+min \{p,q\}-p$, i.e., $z_0$ is not a pole of $\Phi(z)$. \\ $\bullet$ Similarly, $b$-points of $L_{j}f(z)$ are not poles of $\Phi$.\\$\bullet$ Let $z_1$ be a pole of $L_{j}f(z)$ of order $r$. Then, a simple calculation yields that $z_1$ is a simple pole of $\Phi(z)$, regardless of whether it is a pole of $L_{j}f(z+c)$ or not.
Hence, by {\it Lemma \ref{l3.2}}, \beas N(r,\Phi)  \leq \ol N(r,L_{j}f(z))\leq (j+1)\ol N(r,f)=S(r,f).\eeas Further, using {\it Lemmas \ref{l3.1}, \ref{l3.4}} and logarithmic derivative theorem we obtain \beas m(r,\Phi) &\leq& m\left(r,\frac{L_{j}f^\prime(z)}{L_{j}f(z)-a}\right)+m\left(r, \frac{L_{j}f(z)-L_{j}f(z+c)}{L_{j}f(z)-b}\right) \\&\leq& m\left(r,1-\frac{L_{j}f(z+c)-b}{L_{j}f(z)-b}\right)=S(r,f). \eeas
Hence. \bea \label{e4.2}T(r,\Phi)=S(r,f). \eea Now, we denote by $ N_{0}\left(r,\frac{1}{L_{j}f^\prime(z)}\right) $ counting function of those zeros of $L_{j}f^\prime(z)$, which are not zeros of $(L_jf(z)-a)(L_jf(z)-b)$ and similar notation for $L_jf(z+c)$. Now, from (\ref{e4.1}) and (\ref{e4.2}) we have \bea\label{e4.3} N_{0}\left(r,\frac{1}{L_{j}f^\prime(z)}\right) \leq N\left(r, \frac{1}{\Phi}\right)=S(r,f) .\eea  Here, \beas \Phi(z+c)= \frac{L_{j}f^\prime(z+c)(L_{j}f(z+c)-L_{j}f(z+c+c))}{(L_{j}f(z+c)-a)(L_{j}f(z+c)-b)}. \eeas From the above using {\it Lemma \ref{l3.3e}}, we get \bea\label{e4.4} N_{0}\left(r,\frac{1}{L_jf^\prime (z+c)}\right) \leq N\left(r, \frac{1}{\Phi(z+c)}\right)\leq T(r, \Phi(z+c))=S(r,f) . \eea Let us consider a function \bea\label{e4.5} H(z)=\frac{L_{j}f^\prime(z) (L_jf(z+c)-a)(L_jf(z+c)-b)}{L_jf^\prime (z+c)(L_jf(z)-a)(L_jf(z)-b)}.\eea Let $z_0$ be an $a$-point ($b$-point) of $L_jf(z)$ of multiplicity $m$ and that of $L_jf(z+c)$ of multiplicity $n$. Then $z_0$ is a pole of $H(z)$ of order $(m+n-1-m-n+1)=0$, i.e., $z_0$ is not a pole of $H(z)$. Note that, when $m>1$, then zeros of $L_jf(z)$ are zeros of $L_jf^{\prime}(z)$ of order $(m-1)$.\\  Let $z_1$ be a zero of	$L_jf^\prime (z+c)$. Then, the following cases arise: \\ \underline{\em\bf{Case 1:}} Suppose, $z_1$ is a common zero of $(L_jf(z)-a)\; ((L_jf(z)-b))$ and $(L_jf(z+c)-a) ((L_jf(z+c)-b))$.  By the similar arguments that were used before, we can show that $z_1$ is not a pole of $H(z)$. \\ \underline{\em\bf{Case 2:}} $z_1$ is a zero of $(L_jf(z+c)-a)((L_jf(z+c)-b))$ but not of $(L_jf(z)-a)((L_jf(z)-b))$. In that case also, $z_1$ is not a pole of $H(z)$.\\\underline{\em\bf{Case 3:}} If $z_1$ is not zero of $(L_jf(z+c)-a)((L_jf(z+c)-b))$. Then by (\ref{e4.4}), we can say that $z_1$ is not a pole of $H(z)$.  Therefore, combining the above cases we obtain \beas N(r,H) = S(r,f).\eeas Now, using {\it Lemma \ref{l3.1}} and logarithmic derivative theorem we have \beas m(r,H(z)) &\leq & m\left(r, \frac{L_{j}f^\prime(z)}{L_jf^\prime (z+c)}\right)+m\left(r,\frac{L_jf(z+c)-a}{L_jf(z)-a}\right) \\&&+ m\left(r,\frac{L_jf(z+c)-b}{L_jf(z)-b}\right) \\&\leq& S(r,f).\eeas Hence, using the above facts we have \bea\label{e4.6} T(r,H)=S(r,f). \eea \\  We denote by $\ol N_{\times}\left(r, \frac{1}{(L_jf(z+c)-a)(L_jf(z+c)-b)}\right)$, the reduced counting function of zeros of $(L_jf(z+c)-a)\; (\text{resp.}\; (L_jf(z+c)-b))$ which are not the zeros of $(L_jf(z)-a) \;(\text{resp.}\;(L_jf(z)-b))$. By a simple calculation we have, \beas \ol N_{\times}\left(r, \frac{1}{(L_jf(z+c)-a)(L_jf(z+c)-b)}\right) \leq \ol N\left(r, \frac{1}{H(z)}\right) \leq T(r,H)=S(r,f).\eeas Next (\ref{e4.1}) can be rewritten as \bea\label{e4.7} \frac{H(z)L_jf^\prime (z+c)}{(L_jf(z+c)-a)(L_jf(z+c)-b)} = \frac{L_{j}f^\prime(z)}{(L_jf(z)-a)(L_jf(z)-b)}. \eea  Henceforth by $\ol N_{(m,n)}(r,p;L_jf(z),L_jf(z+c))=\ol N_{(m,n)}(r,p),$  $p\in \{a,b\}$; we mean the reduced counting function of those common zeros of $(L_jf(z)-p)$ with multiplicity $m$ and of $(L_jf(z+c)-p)$ with multiplicity $n$.\\ \underline{\em\bf{Claim:}} $\ol N_{(m,n)}(r,p)=S(r,f)$, $p\in \{a,b\}$.
\par Without loss of generality take $p=a$. Suppose on the contrary, there exist $m$, $n$ such that $\ol N_{(m,n)}(r,p)\not=S(r,f)$. Let $z_2 \in \mathbb{C}$ such that $L_jf(z_2)=L_jf(z_2+c)=a$. Then (\ref{e4.7}) gives us, \beas H(z_2)=\frac{m}{n}. \eeas \\ \underline{\em\bf{Case 1:}} $m=n$. Then, $H(z_2)=1$. If $H(z) \not \equiv 1$, then using (\ref{e4.6}) we have \beas \ol N_{(m,n)}(r,a) \leq N\left(r,\frac{1}{H(z)-1}\right)=S(r,f),\eeas which is a contradiction. Hence, $H(z)\equiv 1.$ Therefore, (\ref{e4.7}) can be written as \beas  \frac{L_jf^\prime (z+c)}{(L_jf(z+c)-a)(L_jf(z+c)-b)} = \frac{L_{j}f^\prime(z)}{(L_jf(z)-a)(L_jf(z)-b)}, \eeas which implies that $L_jf(z)$ and $L_jf(z+c)$ share $a$, $b$ and $\infty$ CM and using Theorem B we have $L_jf(z)\equiv L_jf(z+c)$, a contradiction.
\\ \underline{\em\bf{Case 2:}} Suppose that $m \not = n,$  a contradiction. So, there exist $z_3 \in \mathbb{C}$ such that $H(z_3)=\frac{m}{n}$. Using similar arguments as in {\it Case 1}, we obtain $H(z)\equiv\frac{m}{n}$. Then (\ref{e4.7}) can be written as \beas\frac{mL_jf^\prime (z+c)}{(L_jf(z+c)-a)(L_jf(z+c)-b)} = \frac{nL_{j}f^\prime(z)}{(L_jf(z)-a)(L_jf(z)-b)}. \eeas Integrating both sides of the above equation we get \beas \left(\frac{L_jf(z+c)-a}{L_jf(z+c)-b}\right)^m = A \left(\frac{L_jf(z)-a}{L_jf(z)-b}\right)^n,\eeas where $A (\not = 0) \in \mathbb{C}$. Therefore,  by Valiron-Mohon'ko theorem and {\it Lemma \ref{l3.3e}}, $m=n$. \par Hence, the claim is true and $\ol N_{(m,n)}(r,p)=S(r,f),$ for $p \in \{a,b\}$. 
Now, \beas\ol N\left(r,\frac{1}{L_jf(z)-a}\right)&=&\sum_{m,n}\ol N_{(m,n)}(r,a)\\&=& \sum_{m\leq 2,n}\ol N_{(m,n)}(r,a)+\sum_{m \geq 3,n}\ol N_{(m,n)}(r,a)\\&\leq& \frac{1}{3}\sum_{m\geq 3,n}\frac{1}{m}N_{(m}\left(r,a;L_jf(z)\right)\\&  \leq & \frac{1}{3}T(r,L_jf(z))+S(r,f).\eeas Similarly, \beas \ol N\left(r,\frac{1}{L_jf(z)-b}\right) \leq \frac{1}{3}T(r,L_{j}f(z))+S(r,f).  \eeas
  Now, by the Second Fundamental Theorem we get \beas T(r,L_jf(z))& \leq &\ol N(r,L_jf(z))+\ol N\left(r,\frac{1}{L_jf(z)-a}\right)+\ol N\left(r,\frac{1}{L_jf(z)-b}\right)\\&&+S(r,f) \\&\leq& \frac{2}{3}T(r,L_jf(z))+(j+1)\ol N(r,f)+S(r,f),\eeas  which is not possible. Hence, $L_jf(z) \equiv L_jf(z+c)$. 
\end{proof}
\begin{proof}[\underline{\em\bf{Proof of Theorem \ref{t2}}}]
	Set \bea\label{e3.8} \Psi(z) \equiv \frac{ L_{j}f^{(s)}(z+c)}{L_{j}f^{(l)}(z)}.\eea Since, $E(0,L_{j}f^{(l)}(z)) \subseteq E(0,  L_{j}f^{(s)}(z+c))$ and $ E(\infty,  L_{j}f^{(s)}(z+c))\subseteq E(\infty,L_{j}f^{(l)}(z))  $, $\Psi(z)$ is an entire function. Hence, \beas N(r,\Psi(z))=0. \eeas Further, by {\it Lemma \ref{l3.1}, \ref{l3.4e}} and logarithmic derivative theorem,  we know that \beas m(r,\Psi(z)) \leq m\left(r,  \frac{ L_{j}f^{(s)}(z+c)}{L_{j}f^{(s)}(z)} \right) + m\left(r,  \frac{L_{j}f^{(s)}(z)}{L_{j}f^{(l)}(z)} \right)=S(r,f). \eeas Therefore, \beas T(r,\Psi)=S(r,f).\eeas If possible, let $L_{j}f^{(l)}(z) \not \equiv  L_{j}f^{(s)}(z+c).$ Hence, $\Psi(z) \not = 1.$ 
	\\ \underline{\em\bf{Case 1:}} Let $\Psi(z)$ is constant and $z_0$ be an $a$-point of $L_{j}f^{(l)}(z)$. Since,  $ E(a,L_{j}f^{(l)}(z))\\ \subseteq  E(a,  L_{j}f^{(s)}(z+c))$, we can deduce that $\Psi(z_0)=1$, which implies that $\Psi(z)=1,$ which is not possible. Therefore, $a$ is an e.v.p. of $L_{j}f^{(l)}(z)$.
\\ \underline{\em\bf{Case 2:}} $\Psi(z)$ is not constant. If $z_0$ is $a$-point of $L_{j}f^{(l)}(z)$ and $ L_{j}f^{(s)}(z+c)$, by the same argument used as in {\it Case 1}, $\Psi(z_0)=1$. Then by the First Fundamental Theorem, we obtain, \beas  \ol N \left(r, \frac{1}{L_{j}f^{(l)}(z)-a}\right) \leq N\left(r, \frac{1}{\Psi(z)-1}\right)\leq T(r,\Psi)+O(1)=S(r,f).\eeas From (\ref{e3.8}), we have \bea\label{e3.9} \frac{\Psi^{\prime}(z)}{\Psi(z)}=\frac{L_{j}f^{(s+1)}(z+c)}{ L_{j}f^{(s)}(z+c)}-\frac{L_{j}f^{(l+1)}(z)}{L_{j}f^{(l)}(z)}. \eea Let $z_0$ be an $a$-point of $L_{j}f^{(l)}(z)$ of multiplicity $p \geq 2$. Since, $ E(a,L_{j}f^{(l)}(z)) \subseteq  E(a,  L_{j}f^{(s)}(z+c))$, $z_0$ is also an $a$-point of $ L_{j}f^{(s)}(z+c)$ with multiplicity $p \geq 2$. Then, from {(\ref{e3.9})}, it follows that $z_0$ is a zero of $\frac{\Psi^{\prime}(z)}{\Psi(z)}$ of multiplicity $p-1$. Hence, \beas N_{(2} \left(r, \frac{1}{L_{j}f^{(l)}(z)-a}\right)& \leq& 2N\left(r,\frac{1}{\frac{\Psi^{\prime}(z)}{\Psi(z)}}\right) \\& \leq & 2T\left(r,\frac{\Psi^{\prime}(z)}{\Psi(z)}\right) +S(r,f) \\&\leq& 2N\left(r,\frac{\Psi^{\prime}(z)}{\Psi(z)}\right)+S(r,f)=S(r,f). \eeas  Combining the above cases we can say, \bea\label{e3.10} N\left(r, \frac{1}{L_{j}f^{(l)}(z)-a}\right)=S(r,f).\eea Since,  $E(0,L_{j}f^{(l)}(z)) \subseteq E(0,  L_{j}f^{(s)}(z+c))$ and $  E(\infty,  L_{j}f^{(s)}(z+c)) \subseteq E(\infty,L_{j}f^{(l)}(z))$, we deduce that \beas N\left(r,\frac{1}{L_{j}f^{(l)}(z)}\right)-N\left(r,\frac{1}{ L_{j}f^{(s)}(z+c)}\right)\leq 0\eeas and \beas N(r, L_{j}f^{(s)}(z+c))-N(r,L_{j}f^{(l)}(z)) \leq 0.\eeas 
%By {\it Lemma \ref{l3.6}}, we obtain \beas &&m\left(r,\frac{1}{L_{j}f^{(l)}(z)}\right)+m\left(r, \frac{1}{L_{j}f^{(l)}(z)-a}\right) \\&&\leq m(r,L_{j}f^{(l)}(z))+N(r, L_{j}f^{(s)}(z+c) )-N\left(r, \frac{1}{ L_{j}f^{(s)}(z+c)}\right)+S(r,f). \eeas
 Adding $N\left(r,\frac{1}{L_{j}f^{(l)}(z)}\right)+N\left(r, \frac{1}{L_{j}f^{(l)}(z)-a}\right)$ with the both sides of the inequality of {\it Lemma \ref{l3.6}} and applying (\ref{e3.10}) and the First Fundamental Theorem we obtain, \beas 2T(r,L_{j}f^{(l)}(z)) &\leq & N\left(r,\frac{1}{L_{j}f^{(l)}(z)}\right)+N\left(r, \frac{1}{L_{j}f^{(l)}(z)-a}\right)\\&&+m(r,L_{j}f^{(l)}(z))+N(r, L_{j}f^{(s)}(z+c) )\\&&-N\left(r, \frac{1}{ L_{j}f^{(s)}(z+c)}\right)+S(r,f). \eeas i.e., {\small\beas T(r,L_{j}f^{(l)}(z)) &\leq& N\left(r, \frac{1}{L_{j}f^{(l)}(z)-a}\right)+\left(N\left(r,\frac{1}{L_{j}f^{(l)}(z)}\right)-N\left(r, \frac{1}{ L_{j}f^{(s)}(z+c)}\right)\right)\\&&+(N(r, L_{j}f^{(s)}(z+c)))-N(r,L_{j}f^{(l)}(z))+S(r,f) \\&=&S(r,f),\eeas} which is not possible. Therefore, $L_{j}f^{(l)}(z)  \equiv  L_{j}f^{(s)}(z+c).$ 
\end{proof}

\begin{proof}[\underline{\em\bf{Proof of Theorem \ref{t3}}}]
	Let us consider the following auxiliary function \bea\label{e3.11} U(z)= \frac{L_{j}f^{(l+1)}(z)(L_{j}f^{(l)}(z)-L_{j}f^{(s)}(z+c))}{L_{j}f^{(l)}(z)(L_{j}f^{(l)}(z)-a)}.\eea First assume that $U \not \equiv 0.$ \par Let $z_0$ be $a$-point of $L_{j}f^{(l)}(z)$ of multiplicity $p$ and that of $L_{j}f^{(s)}(z+c)$ of multiplicity $q$. Therefore, $z_0$ is a zero of $U$ of multiplicity $(p-1)+\min\{p,q\}-p \geq 0$. Hence, $z_0$ is not a pole of $U$. Further, as $E(0,L_{j}f^{(l)}(z)) \subseteq  E(0,  L_{j}f^{(s)}(z+c))$, if $z_1$ is a zero of $L_{j}f^{(l)}(z)$ of multiplicity $m$, it is a zero of $L_{j}f^{(s)}(z+c)$ of multiplicity same. Therefore, $z_1$ is a zero of $U$ of multiplicity $(m-1)+m-m=m-1 \geq 0.$ Hence, $z_1$ is not a pole of $U$. Next, let $z_2$ be a pole of $L_{j}f^{(l)}(z)$ of multiplicity $n$. By a simple calculation we can say that $z_2$ is a simple pole of $U$. Using {\it Lemma \ref{l3.9}} we have, \beas N(r,U) \leq \ol N (r,L_{j}f^{(l)}(z) )=S(r,f). \eeas Hence, \beas N(r,U)=S(r,f). \eeas Now, using {\it Lemmas \ref{l3.4e}, \ref{l3.5}} and logarithmic derivative theorem we obtain \beas m(r,U) &\leq& m\left(r,\frac{L_{j}f^{(l+1)}(z)}{L_{j}f^{(l)}(z)-a}\right)+m\left(r,1-\frac{L_{j}f^{(s)}(z+c)}{L_{j}f^{(l)}(z)} \right) \\&=&S(r,f).\eeas Therefore, \bea\label{e3.12} T(r,U)=S(r,f).\eea Further, using {\it Lemmas \ref{l3.4e}, \ref{l3.5}} and logarithmic derivative theorem we have \beas&& m\left(r, \frac{1}{L_{j}f^{(l)}(z)}\right) \leq m\left(r, \frac{U}{L_{j}f^{(l)}(z)}\right)+m\left(r,\frac{1}{U}\right)+S(r,f)\\&\leq& m\left(r,\frac{L_{j}f^{(l+1)}(z)}{L_{j}f^{(l)}(z)(L_{j}f^{(l)}(z)-a)}\right)+m\left(r, 1-\frac{L_{j}f^{(s)}(z+c)}{L_{j}f^{(l)}(z)}\right) +S(r,f)\\&=&S(r,f).\eeas Clearly from (\ref{e3.11}) and (\ref{e3.12}) using the First Fundamental Theorem we get \beas N\left(r, \frac{1}{L_{j}f^{(l)}(z)}\right)- \ol N\left(r, \frac{1}{L_{j}f^{(l)}(z)}\right) \leq N\left(r, \frac{1}{U}\right) \leq T(r,U)=S(r,f).\eeas By the hypothesis of Theorem \ref{t3} we get\beas N\left(r, \frac{1}{L_{j}f^{(l)}(z)}\right)=S(r,f).\eeas Hence, \beas T\left(r, \frac{1}{L_{j}f^{(l)}(z)}\right)=S(r,f) ,\eeas which is a contradiction. Therefore, $U \equiv 0$,
i.e., $L_{j}f^{(l)}(z) \equiv L_{j}f^{(s)}(z+c).$	\end{proof}
\begin{proof}[\underline{\em\bf{Proof of Theorem \ref{t4}}}]
	%Since, $E(0,L_{j}f^{(l)}(z)) \subseteq  E(0,  L_{j}f^{(s)}(z+c))$ and $  E(\infty,\\  L_{j}f^{(s)}(z+c)) \subseteq E(\infty,L_{j}f^{(l)}(z)) $, so \bea\label{e3.13}\frac{L_{j}f^{(s)}(z+c)}{L_{j}f^{(l)}(z)}=\gamma(z),\eea where $\gamma(z)$ is a meromorphic function such that $N(r,\gamma(z))=S(r,f).$ Using {\it Lemma \ref{l3.4e}, \ref{l3.5}} we get \beas m(r,\gamma(z))=S(r,f),\eeas which implies that \bea \label{e3.14}T(r,\gamma(z))=S(r,f).\eea
	
	Clearly, $\rho_2(L_jf(z)) \leq \rho_2(f) <1$. 
	%If $\rho_2(L_jf(z))$ is of infinite order, then $L_jf(z)$ can not be a rational function.
	% If $\rho_2(L_jf(z))$ is finite then  
		Proceeding as the same way of {\it Lemma 4 of \cite{Pramanik+Sarkar_Math.Studii_2024}} we can prove that $L_jf(z)$ is transcendental.
	Suppose that, $L_{j}f^{(l)}(z) \not \equiv L_{j}f^{(s)}(z+c),$ i.e., $\Psi (z) \not \equiv 1$, since otherwise we are done. Next considering the auxiliary function (\ref{e3.8}) and using the same arguments as used in the proof of Theorem 
	\ref{t2}, we have \beas T(r, \Psi)=S(r,f).\eeas
	  Let $z_0$ be a common $a$-point of $L_{j}f^{(l)}(z)$ and $L_{j}f^{(s)}(z+c)$. Then, $\Psi(z_0)=1.$\par  Therefore, using the First Fundamental Theorem we have {\small\bea \label{e4.13}\nonumber \ol N\left(r, \frac{1}{L_{j}f^{(l)}(z)-a}\right)&=&\ol N\left(r,\frac{1}{L_{j}f^{(s)}(z+c)-a}\right) \leq N\left(r, \frac{1}{\Psi(z)-1}\right)\\&\leq& T(r,\Psi(z))+S(r,f)=S(r,f).\eea }
	  
	  
	  
	  
%	Set \bea\label{e3.16} F(z)=\frac{L_{j}f^{(l+1)}(z)}{L_{j}f^{(l)}(z)(L_{j}f^{(l)}(z)-a)}-\frac{L_{j}f^{(s+1)}(z+c)}{L_{j}f^{(s)}(z+c)(L_{j}f^{(s)}(z+c)-a)}.\eea From (\ref{e3.16}),  applying {\it Lemma \ref{l3.4e}} and the logarithmic derivative theorem we get \beas m(r, F(z)) &\leq & m\left(r, \frac{L_{j}f^{(l+1)}(z)}{L_{j}f^{(l)}(z)}\right)+ m\left(r, \frac{L_{j}f^{(l+1)}(z)}{L_{j}f^{(l)}(z)-a}\right) \\&&+m\left(r, \frac{L_{j}f^{(s+1)}(z+c)}{L_{j}f^{(s)}(z+c)}\right)+m\left(r, \frac{L_{j}f^{(s+1)}(z+c)}{L_{j}f^{(s)}(z+c)-a}\right)\\&=& S(r,f).\eeas Further, using (\ref{e3.15}), from (\ref{e3.16}) we have, \beas N(r, F(z)) &\leq &\ol N_{*} (r,a;L_{j}f^{(l)}(z),L_{j}f^{(s)}(z+c) )+\ol N_{*} (r,0;L_{j}f^{(l)}(z),L_{j}f^{(s)}(z+c) )\\&\leq& \ol N\left(r, \frac{1}{L_{j}f^{(l)}(z)-a}\right)+\ol N\left(r, \frac{1}{ \Psi(z)}\right)+S(r,f)\\&=& S(r,f).  \eeas Hence, \bea \label{e3.17} T(r,F(z))=S(r,f).\eea\\{\bf{\underline{Case 1:}}} Suppose $F \equiv 0$. Integrating (\ref{e3.16}) we have, \bea\label{e3.18} \frac{L_{j}f^{(l)}(z)-a}{L_{j}f^{(l)}(z)} \equiv A \frac{L_{j}f^{(s)}(z+c)-a}{L_{j}f^{(s)}(z+c)},\eea where $A$ is a non-zero complex number. If $A=1$, we have $L_{j}f^{(l)}(z) \equiv L_{j}f^{(s)}(z+c),$ which is a contradiction. If $A \not = 1,$ from (\ref{e3.18}), a simple calculation yields, \beas L_{j}f^{(l)}(z)+\frac{a}{A-1} \equiv \frac{a}{A-1}\left\{\frac{L_{j}f^{(l)}(z)-a}{L_{j}f^{(s)}(z+c)-a}\right\}. \eeas  Clearly, $-\frac{a}{A-1}\not = 0,a.$ Using (\ref{e3.15}), from the above equation we get, \beas \ol N\left(r, \frac{1}{L_{j}f^{(l)}(z)+\frac{a}{A-1}}\right)=S(r,f). \eeas Using the Second Fundamental Theorem, {\it Lemma \ref{l3.9}} and (\ref{e3.15}) we have, {\small\beas &&T(r,L_{j}f^{(l)}(z))\\& \leq & \ol N\left(r, \frac{1}{L_{j}f^{(l)}(z)-a}\right)+\ol N\left(r, \frac{1}{L_{j}f^{(l)}(z)+\frac{a}{A-1}}\right)+\ol N(r,L_{j}f^{(l)}(z))+S(r,f)\\&=&S(r,f),\eeas} a contradiction. \\{\bf{\underline{Case 2:}}} Next, suppose that $F\not \equiv 0.$
	
	
	
	 From (\ref{e3.8}), we have \beas (L_{j}f^{(s)}(z+c)-a) = \Psi(z)\left( L_{j}f^{(l)}(z)-\frac{a}{\Psi(z)}\right). \eeas
	% we have \beas L_{j}f^{(l)}(z)-a=\frac{1}{\Psi(z)}(L_{j}f^{(s)}(z+c)-a\Psi(z)). \eeas
	 
	 Therefore, by (\ref{e4.13}) we have\beas \ol N\left(r,\frac{1}{L_{j}f^{(l)}(z)-\frac{a}{\Psi(z)}}\right)
	 \leq \ol N\left(r,\frac{1}{L_{j}f^{(s)}(z+c)-a}\right)=S(r,f).\eeas
	 
	
	%Since, $\Psi(z) \not \equiv 1$, there exists a complex number $z_0$ such that $\Psi(z_0)=d \not= 1.$ 
	%\\First suppose that, $a$ is not a Picard's exceptional value of $L_{j}f^{(l)}(z)$. Then from (\ref{e4.17}), we know there exist $z_1$ such that it is an a-point as well as $a\Psi(z_1)$ point of $L_{j}f^{(l)}(z)$.
	% Clearly, from (\ref{e4.17}), zeros of $\left(\Psi(z)-d\right)$ are zeros of $\left(L_{j}f^{(s)}(z+c)-ad\right)$. Hence, we get \beas \ol N\left(r,\frac{1}{L_{j}f^{(s)}(z+c)-ad}\right) =S(r,f).\eeas
	 
	 
	 By {\it Lemma \ref{l3.3e}}, we know $L_jf(z)$ is transcendental and so we have $\Psi(z)$ is small function with respect to $L_jf(z)$, by the Second Fundamental Theorem for small function, (\ref{e4.13}) and the above inequality we have, \beas T(r,L_{j}f^{(l)}(z)) &\leq & \ol N(r, L_{j}f^{(l)}(z))+\ol N\left(r, \frac{1}{L_{j}f^{(l)}(z)-a}\right)\\&&+\ol N\left(r,\frac{1}{L_{j}f^{(l)}(z)-\frac{a}{\Psi(z)}}\right)+S(r,f)\\&=&S(r,f), \eeas which is contradiction. Hence, $L_{j}f^{(l)}(z) \equiv L_{j}f^{(s)}(z+c).$
	\end{proof}
\section{Acknowledgment}
The authors wish to thank the reviewer for his/her valuable suggestions towards the improvement of the paper. 

 \bibliographystyle{amsplain}
	
\end{document}





%------------------------------------------------------------------------------
% End of journal.tex
%------------------------------------------------------------------------------
